Author’s response to reviews

Title: Whole exome sequencing reveals novel COL4A3 and COL4A4 mutations and resolves diagnosis in Chinese families with kidney disease

Authors:

Fujun Lin (seatoneal@hotmail.com)
Fan Bian (bianfan_07@126.com)
Jun Zou (zoujunshcn@hotmail.com)
Xiangru Wu (xiangruwu6@hotmail.com)
Jianping Shan (shanjpjhjd@163.com)
Wei Lu (luweixh@126.com)
Yao Yao (yaoyao19851205@msn.com)
Gengru Jiang (jianggeng-ru@hotmail.com)
Daniel Philip Gale (d.gale@ucl.ac.uk)

Version: 2
Date: 25 July 2014

Author’s response to reviews:

We have made a few changes according to the editorial request on July 24th 2014:

1. Line and page numbers have been included.
2. Full postal address and emails of all submitting authors have been added.
3. A separate conclusions section has been added.
4. In addition, we have noticed 2 minor errors in the labelling of Figure 3A and Figure 4A. Please find the amended figures that are to be used in their places.